SPS operation in Week 33 (from MO 13/08 to MO 20/08) – E. Métral
· MO 13/08/07
- During the morning we had several “30 mm interlock” (MD settup which also gives an interlock for the SFTPRO cycle) => Few minutes of down time.

- 15h54: In TT41, access required through door PPG4115 to repair a BPM, but this door had no key lock, it can only be forced.
- 17h31: End of access to GL300 gallery next to NA48. This zone was on free access, so we made a search patrol after. However the room at the right hand end of the gallery, containing electronics, is locked with a key from the NA48 experiment. We do not have this key, and therefore could not check inside. We checked with NA48 that the room was empty. The TAX02 for P42 line had been set to a reduced hole position, to limit intensity to a safe level for access in GL300 (giving 3 10^11 on T10). We restored the 10mm hole position to go back to nominal intensity, 13 10^11. This is an irregular situation. We should at least have a key to the GL300 electronics room, to make a complete patrol, before restoring full intensity.

- 19h28: 100% of beam lost at the end TT10, right after injection. We checked all elements in TT10, and the MKP, and kick vs BCT on OASIS, all looked ok. We then reloaded the COD settings saved on Saturday evening, and beam injected ok. It looks like a large injection error was introduced by the injection autopilot, which we had running at the time.
· TU 14/08/07: 

- 08h48: No beam from LINAC (cooling water problem in the RF system of the linac) => Beam back at 12h00.
· WE 15/08/07:

- Change to 17 BP (i.e. 20.4 s) supercycle for setting-up of the LHC pilot cycle in the MD segment => LHC pilot (MD2 psb / MD4 ps / MD1 sps) => Cycle called SFT-LHCMD_L14400_PDOT_V1. This SC is a possible option for 2008 for the LHC start-up, allowing permanent beam availability for the LHC but in addition continuous FT beam, so all users do profit. Purpose of the current tests (until end of next week) is to study and optimise the ramp of the LHC pilot beam and see how much beam and intensity can be sent before we miss RF power.
- Problems on SFTPRO => Large oscillations in TT10 and pb of magnetic field at SPS injection. The 2nd pb can be understood in the sense that we have a different MD cycle and therefore different Eddy currents… Solved when Karel decreased the magnetic field by 0.2 A (out of 160 A). The closed orbit was centered in the vacuum chamber. The first pb with the strong kicks at the beginning of TT10 (MAL.1001 + MBIV.1003) is not understood yet.

- The measurement of the radial position along the cycle revealed that the rms radial position drifts from ~ 3 mm to ~ 8 mm between 2000 and 3000 [ms] (~ 2000 ms corresponds to ~ 200 GeV/c) and then come back to ~ 3 mm from 3000 to 4000 [ms]. This problem does not appear on SFTPRO and still has to be understood.
· TH 16/08/07:

- Few stops due to the CPS switching between 3 and 5 current mode.

- Concerning the radial position drift present on the MD cycle and not on SFTPRO, Jorg and Karel had the idea that it could be due to dipolar errors amplified by the 1 / sin [Pi*Q] term. Reminder: the LHC beam uses a low tune Qh =26.13, whereas SFTPRO uses a high tune 26.62. The amplification factor for LHC vs. SFTPRO is ~ 2.3. This has been confirmed on the MD LHC cycle by increasing the horizontal tune to 26.40, which reduced accordingly the orbit deformation.
· FR 17/08/07:

- The MD cycle was used by the BT people to prepare the TI8 tests of next week (August 26).

- Info during the day from J. Uythoven that the TI8 tests foreseen for 26th August and 16th September have been postponed. Depending on the progress of the tests on the LHC equipment installed in 8-1, it should be possible to perform a TI 8 beam test in October or early November, just before the annual shutdown. TI2 tests remain unchanged.

- 17h19: No beam for user SFTPRO1: "chaine 11" (i.e. NorthArea) tripped. It was impossible to do the reset of the “chain 11”. Specialists called and beam back only at 19h22, but problem not yet understood (bad contact?).

· SA 18/08/07:

- 03h51: EA NORTH H2 called: there is a problem with QUAD11. 
We tried to switch it back on, however is stuck to 16.9Amps and it does not go to the beam reference (140Amps). We called the piquet => First line arrived and he could switch on the power converter without any problems (04h12).
· SU 19/08/07:

- 00h58: EA NORTH H2B called: QUAD11 dropped. This means the same power supply (NR11_044) as yesterday => Quad11 OK at 01 h51 after first line intervention (change of a card).

- 04h32: Again NR11_044 in the H2 beamline dropped.

- 09h38: Compensator tripped. TCR calling piquet TS-EL.

- 11h54: chain 11 and 13 tripped => Reason unknown (seems same problem as on Friday) but this time we were able to restart them easily => The specialists will have a look in more detail on Monday morning.

=> Beam back (after pb with the compensator and then chains 11 and 13) at 11h58. The pb. on the compensator is not yet known as there is no apparent reason for that. It seems that they were a slight imbalance between the 3 phases… To be followed-up.

- 12h00: Lots of North area technical gallery access zones fallen to supervised access. Probably due to chain 11 and 13 trip. Zones 811, 814, 815, 818, 842, 854, GL300.

- 12h52: Chain 15 tripped. Enable lamp was flickering but now seems stable (we were able to rearm the chain).

- 12h57: Having trouble restarting P42 Bend5 NR22_026 which is blocking P0 survey => Called firstline for problem on NR22-026.

- 15h26: Call from H6B they have a problem increasing the threshold on their THRES01 when we change the pressure we get an error message.
BI group will look at it on Monday morning. In the meantime the H6 beamline can still use the Cherenkov detector, however without the possibility to change settings.
· MO 20/08/07:

- 03h47: EA NORTH H6: BEND3 - NR22_018 oscillating, the power converter does not give a stable current. We called first line => At 04h39 first line exchanged a control board in the Power converter... OK then, but at 04h46 the same pb came back. The first line was still there, changed another card, which stabilized the current.

- 08h12: No beam for user SFTPRO2. Pulse stop to attempt change to 17BP cycle for MD. Problems with Sequence Manager software. Unable to change due to reservation blocked. Also the sequence manager shows a status as if the 17BP cycle was already running (We have 12BP running). PS call controls expert.
